Author's response to reviews

Title: Illness mapping: A time and cost effective method to estimate healthcare data needed to establish community based health insurance.

Authors:

Erika Binnendijk (erikabinnendijk@gmail.com)
Meenakshi Gautham (Gautham.meenakshi@gmail.com)
Ruth Koren (ruth.ruthkoren@gmail.com)
David M Dror (daviddror@socialre.org)

Version: 2 Date: 17 September 2012

Author's response to reviews: see over
To,

The Editorial Board
BMC Medical Research Methodology

New Delhi, Sept 17, 2012

COVER LETTER

Dear Editors,

On behalf of my co-authors and myself, I am pleased to re-submit the revised version of an original article for publication in your journal. The article entitled **Illness mapping: A time and cost effective method to estimate healthcare data needed to establish community based health insurance**, is based on primary research that we carried out in rural areas of Bihar in India. Pursuant to the comments offered on the original manuscript, we have revised the paper. We now sent both the revised manuscript, and a reply sheet containing detailed responses to all the comments raised by the reviewers and the Editor. The article has been thoroughly reviewed and largely re-written to take account of the comments and suggestions of peer reviewers and the editor. We wish to thank the Editor and the reviewers for their useful suggestions and feedback. We trust that the revised version, submitted herewith, will be acceptable to the Editor and reviewers, and we look forward to hearing from you soon. Thanks in advance, and best regards,

Sincerely,

David M Dror

Corresponding author.